Author's response to reviews

Title: A Simplified Guide Ruler for Rotational Osteotomy

Authors:

Chen-Kun Liaw (sampliu@ha.mc.ntu.edu.tw)
Rong-Sen Yang (rsyang@ntuh.gov.tw)
Sheng-Mou Hou (shengmou@ntuh.gov.tw)
Tai-Yin Wu (dienewu@yahoo.com.tw)
Chiou-Shann Fuh (d92008@csie.ntu.edu.tw)

Version: 4 Date: 14 December 2007

Author's response to reviews: see over
Dear Editor:

Followed your suggestion, we revised our manuscript. It is a minor revision, we only add some subheading so that the readers can understand it more. These are list below:

Line 41: Background of Rotational Osteotomy
Line 53: Report of Ćobeljić et al.
Line 65: Mechanism of Method of Ćobeljić et al
Line 74: Our Improvement
Line 88: Our Guide Ruler
Line 100: Verification of Our Guide Ruler
Line 135: Different Strategies of Fixating Osteotomy
Line 156: List of abbreviations
Line 157: SD: standard deviation
Line 158: NSC: National Science Council

Corresponding Author
Chiou-Shann Fuh